Author's response to reviews

Title: The relationship between baseline Organizational Readiness to Change Assessment subscale scores and implementation of hepatitis prevention services in substance use disorders treatment clinics.

Authors:

Hildi J Hagedorn (hildi.hagedorn@va.gov)
Paul W Heideman (paulwheideman@gmail.com)

Version: 6 Date: 7 June 2010

Author's response to reviews: see over
June 7, 2010

To: Implementation Science Editorial Team

I have completed the review of changes made to my manuscript. There was one instance where I did not accept a change. On page 10, four lines from the bottom, you had changed the word "which" to "where". However, I think the appropriate word that I intended was "with".

As advised, I moved Table 1 to a separate document and I reformatted Tables 2 and 4 as Table Objects.

Thank you for your assistance with my manuscript,

Hildi J. Hagedorn, PhD, LP
Implementation Research Coordinator
Substance Use Disorders QUERI
Minneapolis VA Medical Center
One Veterans Drive (116A9)
Minneapolis, MN 55417
Phone: 612-467-3875
Email: hildi.hagedorn@va.gov